Author's response to reviews

Title: Comparison of ICD code-based diagnosis of obesity with measured obesity in children and the implications for health care cost estimates

Authors:

Stefan Kuhle (stefan.kuhle@ualberta.ca)
Sara FL Kirk (sara.kirk@dal.ca)
Arto Ohinmaa (arto.ohinmaa@ualberta.ca)
Paul J Veugelers (paul.veugelers@ualberta.ca)

Version: 3 Date: 14 December 2011

Author's response to reviews:

Dear Hayley,

We have fixed the sentence in the Discussion as suggested by Reviewer 1. Upon checking the manuscript again for compliance with the journal guidelines, we found no issues.

We did one minor change regarding the BMI measurement in the Methods: It now reads "[...] in light indoor clothing" instead of "with full clothing" [which may falsely suggest overcoats, hats etc].

Kind regards,

Stefan Kuhle (on behalf of Paul Veugelers)